Author’s response to reviews

Title: Early screening the small bowel is key to protect Peutz-Jeghers Syndrome patients from surgery: a novel mutation c.243delG in STK11 gene

Authors:

Yu-Liang JIANG (sting_321@163.com)
Zi-Ye ZHAO (yemenzhao@126.com)
Bai-Rong LI (airbai8040@163.com)
Jing LI (lijing@yeah.net)
Xiao-Wei JIN (jinxiaowei@yeah.net)
En-Da YU (endayu@yeah.net)
Xiao-Dong XU (xiao_dong214@126.com)
Shou-Bin NING (ningshoubin@126.com)

Version: 4 Date: 11 Feb 2019

Author’s response to reviews:

ID: BMGE-D-17-00567R3

Title: Early screening the small bowel is key to protect Peutz-Jeghers Syndrome patients from surgery: report of a novel STK11 mutation (c.243delG)

Point-by-point response:

1) Overlap.

Response: We have rephrased the manuscript.

2) Funding.

Response: We have added the corresponding descriptions.

3) English.
Response: We have rephrased the manuscript with the help of our native English speaking colleague.

4) Clean Manuscript.

Response: Done.

Other changes:

1. We have modified the figure.